

Ahmed, Yousry and Elshandidy, Tamer (2016) The effect of bidder conservatism on M&A decisions: text-based evidence from US 10-K filings. International Review of Financial Analysis, 46 . pp. 176-190. ISSN 1057-5219

#### Access from the University of Nottingham repository:

http://eprints.nottingham.ac.uk/43020/1/the%20effect%20of%20bidder%20conservation %20on%20M%20A%20decisions.pdf

#### Copyright and reuse:

The Nottingham ePrints service makes this work by researchers of the University of Nottingham available open access under the following conditions.

This article is made available under the Creative Commons Attribution Non-commercial No Derivatives licence and may be reused according to the conditions of the licence. For more details see: http://creativecommons.org/licenses/by-nc-nd/2.5/

#### A note on versions:

The version presented here may differ from the published version or from the version of record. If you wish to cite this item you are advised to consult the publisher's version. Please see the repository url above for details on accessing the published version and note that access may require a subscription.

For more information, please contact eprints@nottingham.ac.uk

### The effect of bidder conservatism on M&A decisions: Text-based evidence from US 10-K filings\*

Yousry Ahmed

University of Bristol, UK Yousry.ahmed@bristol.ac.uk

Tamer Elshandidy\*\*

University of Bristol, UK

Tamer.Elshandidy@bristol.ac.uk

\*\*Correspondence should be addressed to Tamer Elshandidy, School of Economics, Finance and Management, University of Bristol, 8 Woodland Road, Clifton, Bristol, BS8 1TN. Email: <u>Tamer.Elshandidy@bristol.ac.uk</u>

<sup>\*</sup> We are grateful to Professor Brian M Lucey (The Editor) and the two anonymous referees for helpful comments and suggestions. We would also like to thank Kenneth R. Ahern, Margaret Greenwood, Semi Kedia, Sandy Klasa, Gordon Phillips, Tim Loughran, Nick Taylor, Fanis Tsoligkas, Jon Tucker, Steven Young and Osman Yukselturk, and participants at the PhD Conference in Monetary and Financial Economics at University of the West of England Bristol, the BAFA doctoral conference and the SWAG BAFA conference for their useful comments and constructive suggestions. Ahmed appreciatively acknowledges the financial support of Zagazig University in Egypt for funding this study.

## The effect of bidder conservatism on M&A decisions: Text-based evidence from US 10-K filings

#### Abstract:

This paper examines whether and how bidders' conservative tone in 10-K filings influences the subsequent merger and acquisition (M&A) investment decisions of these US firms from 1996 to 2013. Based on 39,260 firm-year observations, we find, consistent with behavioural consistency theory, that conservative bidders are less likely to engage in M&A deals. Further, those that decide to engage in M&As are likely to acquire public targets and within-industry firms. These bidders are inclined to employ more stock acquisitions than cash acquisitions. Our results also indicate that conservative bidders experience abnormally poor stock returns around the announcements of M&A investments. This provides new insights on the mechanism through which bidders' sentiments influence shareholders' wealth. Overall, these findings highlight the implications of the textual sentiment of corporate disclosure for the forecasting of corporate investment and financing decisions. Our results have practical implications, since they shed light on the value relevance of the information content of major Securities Exchange Commission (SEC)-mandated 10-K filings.

**Keywords:** Textual sentiment, Bidders' conservative tone, Behavioural consistency theory, Risk averse, Payment method, Market reaction, Types of acquisitions.

JEL classification: G02, G32, G34

#### **1. Introduction**

The textual analysis of corporate disclosure provides insights into managerial expectations and identifies important qualifiers and caveats that are absent from financial measures (Loughran & McDonald, 2015a). Barber and Odean (2008) and Engelberg and Parsons (2011) document that the sentiment of qualitative corporate news drives market trading. Feldman, Govindaraj, Livnat, and Segal (2010) find that the disclosure tone of the 10-K has incremental information content beyond that of financial measures. Loughran and McDonald (2011) observe that firms that convey more negative tones in their 10-K filings experience negative market reactions around the filing dates. Arnold, Fishe, and North (2010) quantify the words used in the risk factors sections of IPO prospectuses and find that they capture most of the IPO returns. Yan (2015) shows that the tone of 10-Ks gives accurate predictions of firms' long-term performance. Thus, extant literature suggests that words selected by managers in textual corporate communications have significant predictive power for firms' future stock returns, earnings, and performance. However, relatively little is known about whether the sentiment of corporate disclosure could be employed to anticipate investment activity. This paper addresses this gap by investigating the impact of textual sentiment on corporate merger and acquisition (M&A) investment decisions.<sup>1</sup>

Further, Ferris, Hao, and Liao (2013) introduce the concept of issuer conservatism, measured as the proportion of negative words in the IPO prospectus. They find that issuer conservatism is inversely related to subsequent operating and stock return performance.<sup>2</sup> We adopt the same proxy of the proportion of negative tone in the 10-K to examine the effect of

<sup>&</sup>lt;sup>1</sup> Kearney and Liu (2014) argue that the textual sentiment of corporate disclosure provides a good reflection of firms' internal and external conditions and thus has significant power to explain subsequent corporate investment decisions.

 $<sup>^2</sup>$  In line with that, Tetlock, Saar-Tsechansky, and Macskassy (2008) find that the proportion of negative words in firms' specific news stories forecast firms' future earnings and returns above and beyond stock analysts' forecasts and historical accounting data.

textual conservatism on M&A decisions. M&A deals constitute one of the most strategic investment decisions, with substantial effects on firms' current and future value (Custódio & Metzger, 2013). The 10-K is a major and critical public document that provides a comprehensive review of corporate policies and practices. It is also more informative than any other business communication (Loughran & McDonald, 2014). Consequently, this paper examines the impact of cautionary language in 10-K reports (henceforth bidder conservatism) on subsequent M&A decisions.<sup>3</sup>

Cronqvist, Makhija, and Yonker (2012) argue that, consistent with the notion of the endogenous matching of managers with firms' mechanisms, firms choose managers with characteristics that fit their own beliefs and behaviours. They find that CEOs who are more conservative in terms of their personal leverage are optimal at managing firms with more conservative debt policies. They further observe that a firm may even replace a CEO with another who has the same attributes and behaviours. Accordingly, examining the information content of the 10-K provides a strong understanding of board, manager and firm-matched traits and their implications for future investment decisions. In view of the nature of M&A deals, which require intense board-manager discussions in the bid for final approval (Levi & Zhang, 2014), such deals constitute an ideal platform on which to test the ex-ante impact of bidder conservatism on M&A investment decisions. These M&A investment decisions include the likelihood of making an acquisition, the type and size of acquisitions (public vs. non-public, within-industry vs. cross-industry targets), the payment method, and how the market perceives those investment decisions.

Behavioural consistency theory states that the behaviour of an individual in one situation could forecast his/her future behaviours in other situations (Funder & Colvin, 1991).

<sup>&</sup>lt;sup>3</sup> We focus on the negative tone of firms' 10-K filings, since our main independent variable is bidder conservatism, which is proxied by the ratio of negative words to total words, as in Ferris et al. (2013).

Motivated by this theory, we posit that managers who convey a conservative tone in their 10-K filings can be expected to later make conservative corporate decisions. Conservative behaviour is defined as opposition to change and a tendency toward risk aversion (Wilson, 2013). On the other hand, M&As are viewed as risky investments due to the uncertainty associated with forecasting their outcomes (Levi & Zhang, 2014).<sup>4</sup> Thus, we expect that bidders with a prior conservative tone will be less likely to engage in M&A activities. Further, bidders might have better access to the information needed to estimate the real value of public targets than that required for non-public targets (Capron & Shen, 2007). Consequently, we posit that conservative bidders will want to avoid the uncertainty and asymmetry of information related to acquiring non-public targets. Similarly, we expect that conservative bidders will prefer to acquire familiar and same-industry targets rather than risky industrially diversified acquisitions.<sup>5</sup> Eckbo (2009) documents that bidders prefer stock financing if there is more uncertainty about the target. Consequently, we conjecture that conservative bidders are likely to engage in stock acquisition deals. Prior literature (i.e., Tetlock, 2007; Ferris et al., 2013; Loughran & McDonald, 2011) documents that the market reacts negatively to a conservative tone in different business communications. Accordingly, we expect that conservative bidders will experience negative market reactions around the announcements of M&A deals.

Based on 39,260 firm-year observations over the period between January 1996 and December 2013, we find significant evidence that firms with a more negative tone in their prior 10-Ks are less likely to be involved in subsequent acquisitions and have a small portion of M&A investments. This confirms our conjecture that conservative behaviour is related to risk-

<sup>&</sup>lt;sup>4</sup> Potential outcomes of M&As might be positive, negative or even losses that exceed the initial investment value (Christensen, Dhaliwal, Boivie, & Graffin, 2014). For instance, Microsoft Corporation experienced a loss of \$12.6 billion in their market value after announcing they had acquired Nokia's mobile phone business for \$7.2 billion (Amihud, DeLong, & Golubov, 2013).

<sup>&</sup>lt;sup>5</sup> Berger and Ofek (1995) report that, in 1995, multiple-segment firms with valuation discounts experienced total value losses of \$800 billion. This issue is known as a diversification discount problem.

averse attitudes and a lower tendency to invest in risky investments such as M&As. We further find that, conditional on the making of acquisitions, conservative bidders are more willing to acquire public targets. This might be due to information asymmetry between public and nonpublic firms (Capron & Shen, 2007). In addition, these firms prefer to pursue within-industry acquisitions. This finding is consistent with the proposition that corporate conservatism is more inclined towards familiar and safe than unfamiliar stimuli (Glasgow, Cartier, & Wilson, 1985). Our results also indicate that conservative bidders prefer stock acquisitions over cash acquisitions. Finally, we find that these firms experience a negative market reaction after engaging in these kinds of investments. This supports the notion that the sentiment of qualitative corporate reports drives market trading (e.g., Barber & Odean, 2008; Engelberg & Parsons, 2011).

Our paper contributes to the area of the information content and value relevance of Securities Exchange Commission (SEC)-mandated filings, the 10-K. To the best of our knowledge, no previous study has explored the link between the sentiment of corporate disclosure and subsequent corporate investment decisions in the context of M&As. We provide novel evidence that cautionary language plays a significant role in predicting the likelihood of a firm making an acquisition, and its value. These findings support the prior premises that conservative behaviour is associated with risk aversion and preserving the status quo. Similarly, we add to the literature by showing that the textual analysis of 10-Ks provides significant predictive power for not only the likelihood, but also the type and size, of subsequent acquisitions. These findings confirm the ability of the non-financial information revealed in the narrative sections of 10-Ks to forecast managers' decisions regarding M&As. They also add to the emerging evidence on the usefulness of SEC filings, including 10-Ks (Loughran & McDonald, 2011, 2014, 2015a).

We contribute to a growing literature that uses text analysis in finance, by investigating the implications of bidders' pre-textual sentiment for the anticipation of sources of finance in M&A deals. We observe that bidders that use a conservative tone in their corporate disclosure also have conservative financing policies. Our findings provide new evidence to support the proposition of behavioural consistency theory that individuals behave in a consistent way across different situations (Funder & Colvin, 1991). Further, this paper provides novel evidence on the interdependence between corporate disclosure and shareholders' wealth. We find evidence that soft information embedded in the descriptive sections of the 10-Ks plays a significant role in how the market evaluates firms' stock returns after they have engaged in M&A events. Collectively, these results confirm that a conservative tone in bidders' 10-Ks conveys information that is economically relevant to the prediction of corporate M&A decisions.

Our findings have several implications for both theory and practice in the area of corporate disclosure and behavioural finance. First, with regard to theory, our results provide some insights on the implications of bidders' sentiment and attitude for future M&A investment choices. Our findings reveal that investors and stakeholders might not be able to fully grasp corporate policies and decisions unless they carefully analyse the sentiment of the narrative sections of the 10-K. This suggests the interesting possibility that major corporate investment decisions are derived from more than traditional economic trade-offs and incentives. It also confirms that the textual sentiment of the 10-K enables us to gain an in-depth understanding of firms' conditions, private information, managers' incentives, and expected behaviours by seeing the world through their eyes. Besides this, managers should comprehend the mechanism through which their corporate communications influence their firms' value and shareholders' wealth. They should assess their prior negative disclosure tone before engaging in M&A deals, since it will have a significant effect on how the market will perceive the deals. Second, for

practice, our findings present new insights for policy makers, showing the incremental valuation-relevance of SEC filings and mandatory disclosures, since the textual sentiment of the 10-K could be used to predict corporate investment actions and their consequences.

The remainder of the paper is organised as follows. Section 2 presents our main hypotheses. Section 3 explains our methodology and empirical models. Section 4 introduces our empirical results. Section 5 reports some robustness checks. Section 6 presents our main conclusions.

#### 2. Hypothesis development

#### 2.1. Bidder conservatism and the likelihood of making an acquisition

Behavioural consistency theory advocates the notion that individuals behave in a consistent way across different situations (Funder & Colvin, 1991). Empirically, Cronqvist et al. (2012) show that managers exhibit consistent behaviour in their personal and corporate leverage policies. Motivated by this theory, we conjecture that bidders that use a conservative tone in their corporate disclosure can be expected to make conservative investment decisions. Conservative behaviour is also associated with a high tendency for uncertainty avoidance and preserving the status quo (Jost, Glaser, Kruglanski, & Sulloway, 2003). According to Schneider and Lopes (1986), individuals can be classified according to their attitude towards risk, into the risk-averse and the risk-seekers. The risk-averse group is driven by a desire for security or avoiding negative outcomes, while the risk-seekers group is driven by a desire for high returns. Using a survey experiment on a sample of 761 individuals, Kam and Simas (2010) observe that individuals with highly risk-averse attitudes have a greater preference for policies with certain outcomes than do individuals with high risk tolerance.

On the other hand, M&A deals are risky investments, since it is hard to predict their outcomes due to their uniqueness, scope, size, value and other factors that will affect the bidders' overall performance (Levi & Zhang, 2014). Bidders might thus experience positive

outcomes, negative outcomes, or even losses that exceed the initial value of the M&A deal (Christensen et al., 2014). For instance, Rakuten experienced a drop of \$2 billion in stock value that exceeded the original \$900 million acquisition value of the Viber internet messaging corporation (Yasu & Mukai, 2014). Therefore, motivated by behavioural consistency theory, we expect firms that convey a conservative tone in their 10-Ks are less likely to be involved in subsequent M&A deals. This discussion leads to the following hypothesis:

*H1.* There is a negative relationship between bidder conservatism and the likelihood of making an acquisition.

#### 2.2 Bidder conservatism and type of acquisition

In the previous section we proposed that conservative bidders might prefer their current status quo and would therefore make fewer M&A transactions. In this section, we review the literature to find out, in the case of conservative bidders that do decide to make M&A deals, which type of M&A they might prefer.

The first aspect of acquisition type is whether the acquired firm is a public or a nonpublic firm. According to Jost et al. (2003), conservative individuals exhibit a strong disposition towards preserving the status quo and avoiding uncertainty. They also prefer safe and traditional forms of organisation (Wilson, 2013). Conservative bidders' choice between public and non-public targets could be explained by the difference in information availability for those types of target. Officer (2007, p.573) states, *"While information asymmetry is endemic to all mergers or acquisitions, this problem is likely to be most severe ... [for unlisted targets], in which standards for information disclosure are not as high as for publicly traded firms and information about subsidiaries may be obscured by the parent's financial reporting choices".* Thus, managers of non-public firms typically have better control over the information they want to communicate (Reuer & Ragozzino, 2008). The relaxed disclosure requirements for non-public firms increase the uncertainty about their value relative to public firms (Officer, Poulsen, & Stegemoller, 2009). In contrast, public firms are already priced by the markets and are subject to mandatory disclosure requirements and analysts' coverage (Capron & Shen, 2007). Thus, we presume that a US-quoted conservative bidder might acquire public targets to avoid the increased uncertainty and asymmetric information associated with acquiring non-public targets.<sup>6</sup> This discussion leads to the following hypothesis:

*H2.* There is a positive relationship between bidder conservatism and the likelihood of making a public acquisition.

The second aspect of acquisition type is whether the target comes from the same industry as the bidder or from a different one. If the target firm operates outside the bidder's industrial segment, the acquisition deal is viewed as an industrially diversified acquisition (Duchin, 2010). Previous studies (i.e., Lang & Stulz, 1994; Lins & Servaes, 1999; Lamont & Polk, 2001) document the negative influence of diversified acquisitions on firms' value. Berger and Ofek (1995) find that, during 1986-1991, US conglomerate firms operating in multiple business segments are priced at a mean discount of about 13% to 15% below the sum of the imputed values of their segments. This is known as the diversification discount. Morck, Shleifer, and Vishny (1990) observe that bidders that acquire targets from a different industry experience negative returns. According to Wilson (2013), a conservative attitude is viewed as opposition to change and a tendency toward risk aversion. Glasgow et al. (1985) note that a conservative person is more inclined toward familiar and safe stimuli than unfamiliar stimuli.

<sup>&</sup>lt;sup>6</sup> This conjecture does not contradict previous empirical evidence (e.g., Fuller, Netter, & Stegemoller, 2002; Ang & Kohers, 2001; Moeller, Schlingemann, & Stulz, 2004) that, unlike public acquisitions, non-public acquisitions achieve significant gains around bid announcements and thus could be viewed as less risky investments. Ekkayokkaya, Holmes, and Paudyal (2009) find the superior gains from non-public acquisitions in the short run to be attributable to a lack of information about their value. They also show that those non-public deals suffer a subsequent drop in stock returns in the long run, while the returns on public acquisitions break even over periods of up to three years. These findings support the view that limited information at bid announcements leads the market to overestimate the value effects of non-public acquisitions. However, in the long run, when better-quality information becomes available, the market corrects for its initial overestimation and values non-public acquisition deals downwards. In unreported tables, we run an event study after making a distinction between a short window (3 days around the announcement date) and long windows (12, 24, 36, 48, and 60 months), similarly to Ekkayokkaya et al. (2009), and obtain consistent results.

They have greater sensitivity to the possibility of loss and are more cautious about making major changes (Jost et al., 2003). Consequently, we expect that conservative bidders would prefer to acquire familiar and same-industry targets rather than industrially diversified ones. This discussion leads to the following hypothesis:

*H3.* There is a negative relationship between bidder conservatism and the likelihood of making an industrially diversified acquisition.

#### 2.3. Bidder conservatism and M&A payment method

Bharadwaj and Shivdasani (2003) observe that debt issuance is the main way of raising the cash required to pursue M&A deals. Hutton, Jiang, and Kumar (2014) find that managers who adopt conservative ideologies are more likely to take more conservative financial decisions, such as lower leverage ratios. Motivated by behavioural consistency theory, we presume that bidders that use cautionary language in their 10-Ks are likely to maintain a conservative debt level. Thus, they are likely to finance their acquisitions using their own stock. In addition, Martin (1996) shows that stock acquisition is preferable when the acquiring firm considers the target a risky investment. According to our previous assumptions, a conservative bidder might view M&A deals as risky investment decisions due to the increased variance in their expected outcomes (Christensen et al., 2014). Collectively, conservative bidders might have conservative debt policies and consider M&A transactions unsafe investments. Thus, we expect that conservative bidders are likely to rely more on their own equity to finance M&A deals. This discussion leads to the following hypothesis:

*H4.* There is a positive relationship between bidder conservatism and the likelihood of making a stock acquisition deal.

#### 2.4. Bidder conservatism and market reaction

Previous studies (e.g., Baker & Wurgler, 2006; Tetlock et al., 2008) find that soft information embedded in corporate disclosure has a superior impact on the market's evaluation of firms' stock returns than quantitative measures of firms' fundamentals. Tetlock (2007) finds that the proportion of negative words of firm-specific news significantly explains the movement in the stock market. Ferris et al. (2013) show that the market reacts negatively to IPO prospectuses when the tone of the issuer is conservative. Arnold et al. (2010) quantify the words used in the risk factors section of the IPO prospectus and find that they explain most of the initial and subsequent stock returns. Loughran and McDonald (2011) find that a conservative tone in 10-K filings has a negative effect on firms' stock returns around the filing dates. These studies confirm that firms' stock market prices efficiently incorporate the linguistic information disclosed in corporate reporting. They also show that firms that convey a conservative tone experience poor market reactions. Thus, we expect that the market will react negatively to announcements of M&A transactions by conservative bidders. This discussion leads to the following hypothesis:

*H5.* There is a negative relationship between bidder conservatism and market reactions around the announcements of M&A deals.

#### 3. Methodology

#### 3.1. Sample

We collect annual 10-K filings of US publicly traded firms from the Electronic Data Gathering, Analysis, and Retrieval (EDGAR) database maintained by the SEC. 10-K filings have a similar basic structure across most firms, as regulated by the SEC. Thus, we exclude amended filings from our sample (Balvers, Gaski, & McDonald, 2015).<sup>7</sup> We further eliminate HTML, encoded images, and tables from each 10-K file, since we are interested in the narrative

<sup>&</sup>lt;sup>7</sup> Following Loughran and McDonald (2014), if the 10-K is not available, we use some other type of 10-K such as the 10-K405, 10-KSB, or 10-KSB405. These types represent 9% of our sample. A 10-K405 filing is similar to the 10-K except that it indicates that the CEO did not disclose insider trading activities in time. This form was abolished at the end of 2002. A 10-KSB is an annual report filing designed for small firms and was abolished at the end of 2002. A 10-KSB405 is a 10-K405 for small US firms and was abolished at the end of 2002.

sections of these reports (Loughran & McDonald, 2014). From May 1996 onwards, all US public firms have been required by the SEC to disclose 10-K reports electronically.<sup>8</sup> Our main independent variables are measured in the fiscal year preceding the announcements of M&A deals. Thus, our 10-K sample covers the period between January 1996 and December 2013.

We then link the 10-K documents with the Compustat and CRSP databases using different identifiers, including the Central Index Key (CIK) number associated with each filing, the Compustat GVKEY and the CRSP PERMNO.<sup>9</sup> This merging reduces the original sample of 10-Ks from 67,681 to 46,410 observations during our sample period. We eliminate financial firms (SIC 6000–6999) and utilities (SIC 4900–4999) from the sample, since words such as risk and causality that have negative connotations for non-financial firms might not be perceived in the same way for financial firms (Jegadeesh & Wu, 2013). Additionally, these firms are subject to regulatory constraints and different accounting considerations (Uysal, 2011). We omit firm-year observations with missing data for any variable that we use in our empirical models during the sample period. The above steps lead to a sample of 39,260 firm-year observations.<sup>10</sup>

From the Thomson One database, we download M&A data for non-financial US public acquirers for the period between 1st January 1997 and 31st December 2014. Following Uysal (2011) and Karampatsas, Petmezas, and Travlos (2014), to be included in our sample a target may be a public, a private or a subsidiary firm. Acquisition deals should be completed and may be either domestic or foreign. The payment method has to be cash, stock, or a combination of

<sup>&</sup>lt;sup>8</sup> Electronic filings of 10-Ks were phased in over a three-year period prior to being made compulsory in May 1996. Thus, 10-K filings during those three years were biased toward large firms. We obtain the same results when we start our sample in 1993.

<sup>&</sup>lt;sup>9</sup> We thank Bill McDonald for affording us access to his data file so that we could create a list of CIK numbers and filing dates for our sample of 10-K reports.

<sup>&</sup>lt;sup>10</sup> Loughran and McDonald (2014) keep only those 10-K filings with at least 2,000 words. Lehavy, Li, and Merkley (2011) eliminate filings with fewer than 3,000 words to minimise errors arising from filing transmission or incomplete filing. It is notable that excluding those files from our sample does not affect the results.

both. The acquisition deal represents at least 50% of the target. We exclude all transactions labelled as minority stake purchases, acquisitions of remaining interest, privatisations, leveraged buyouts, spinoffs, recapitalisations, self-tenders or exchange offers, and repurchases (Alexandridis, Fuller, Terhaar, & Travlos, 2013). We employ, as a cut-off point, a minimum deal value of \$1 million in order to avoid obtaining results generated by very small targets (Uysal, 2011).

#### 3.2. Bidder's conservative tone

Following Ferris et al. (2013), we measure conservative tone as the number of negative words divided by the total number of words in the narrative sections of the 10-K filings, multiplied by 100. We employ Loughran and McDonald's (2011) negative word list, since it is more relevant in a business communication context than other lists (e.g., Harvard's General Inquirer (GI); Diction; Henry, 2008). Harvard GI and Diction's negative word lists are drawn from a broader social science context and thus some of their entries might not have negative connotations in business disclosure (Kearney & Liu, 2014). Loughran and McDonald (2011) report that almost 75% of the Harvard GI's negative words are inappropriate for capturing a negative tone in business applications. Harvard words such as "tax, cost, capital, board, and depreciation" do not have a negative meaning when used in 10-K reports (Kearney & Liu, 2014). Diction has 920 pessimistic words that can be obtained by combining the three subcategories of blame, hardship, and denial. Using a sample of 77,158 10-Ks, Loughran and McDonald (2015b) report that 70% of Diction's negative words are misclassified and do not generally indicate a pessimistic tone. For instance, the word "vice" is one of Diction's negative words that does not have the same negative effect in regards corporate textual information (Ferris et al., 2013). Henry (2008) introduces a negative list that contains 85 words developed from corporate earnings press releases. The main drawback is the limited number of words included in this list, which means that it misses important negative words such as "loss, losses,

adverse and impairment" (Loughran & McDonald, 2015b). Consequently, we use Loughran and McDonald's (2011) comprehensive negative world list extracted from a large sample of 10-K reports and created specifically for analysing corporate documents.

We also employ a negative word list only, and not a positive one, since we are interested in evaluating the issue of bidder's conservatism. Furthermore, negative words are clearer than positive words. Managers will rarely negate negative words in order to deliver a positive statement. Conversely, many positive words may be employed to make a negative statement (Loughran & McDonald, 2011). Tetlock et al. (2008), Loughran and McDonald (2015a), and Garcia (2013) clarify that a negative word list provides more incremental information and has significantly better explanatory power than a positive list. Loughran and McDonald's (2011) negative list contains 2,329 words commonly used by company managers. Several studies confirm that this negative word list is appropriate for gauging tone in different business contexts, for instance, for analysing the sentiment of the MD&A section of the 10-K report (Feldman et al., 2010), analyst reports (Huang, Zang, & Zheng, 2014), earnings conference calls (Matsumoto, Pronk, & Roelofsen, 2011), and the IPO prospectus (Ferris et al., 2013; Loughran & McDonald, 2013).

Our measure of bidder conservatism using the ratio of negative words to total words has several advantages over a manual scoring of the 10-K documents. It avoids potential misinterpretations or inconsistencies that are associated with a subjective categorisation of 10-Ks (Li, 2010). It is a continuous variable that improves the power of our empirical tests in comparison to the subjective dummy variables used in prior literature (e.g., Christensen et al., 2014; Hutton et al., 2014). It is also divided by the total number of words in the 10-K, which enables comparisons to be made between 10-Ks of different lengths. It is objective and could easily be replicated for large quantities of corporate disclosures, through the power of computer analysis (Rogers, Buskirk, & Zechman, 2011). In contrast, manual content analysis has a small sample size that affects the power of the analysis and the generalisability of the results (Li, 2010).

Further, our word-list methodology might be better than a complicated technique such as the naïve Bayesian algorithm. The naïve Bayesian methodology is always viewed as a "black box procedure", since little is known about the rules and filters used to analyse the contexts of documents (Loughran & McDonald, 2015a). Usually, a small sample of documents is used to train the program. Then, the naïve Bayesian algorithm starts to classify the tone of the documents according to unseen classification filters (Buehlmaier & Whited, 2014; Antweiler & Frank, 2004). We cannot tell which sentences are classified as positive, negative and neutral, or which phrases or words dominate the analysis (Loughran & McDonald, 2015a). In some cases, texts are manually coded by non-native speakers (see Li, 2010; Huang et al., 2014). Thus, it is hard to replicate the results or test their reliability. Finally, the naïve Bayesian methodology is more time consuming and costly than the word-list approach (Kearney & Liu, 2014).

#### 3.3. Empirical models

We use the estimated prior conservatism of the bidder to investigate its impact on the subsequent M&A investment decisions and economic consequences. First, we combine M&A data with all US public firms from 1996 to 2013 to create a variable that represents the likelihood of a firm undertaking an acquisition. Then, we employ the following probit model to test H1, which addresses the effect of bidder conservatism on the likelihood of pursuing an acquisition:

P (Acquisition=1) =  $\Phi$  ( $\beta_0 + \beta_1$  Bidder conservatism <sub>i,t-1</sub> +  $\sum \beta_i$  Controls <sub>i,t-1</sub>) (1)

Here, Acquisition is a dummy variable that takes the value one if the firm makes an acquisition and zero otherwise. Bidder conservatism is the negative word count scaled by the total number of words in the 10-K with a filing date preceding the announcement date of an

acquisition. To control for firm characteristics, we include several variables based on prior literature. Uysal (2011) finds that firms with higher profitability ratios have a greater ability to acquire other firms. The profitability ratio is measured as earnings before interest, tax and depreciation (EBITD) divided by total assets, at the fiscal year-end preceding the acquisition (Harford, 1999). Uysal (2011) shows that firms with high leverage ratios are less likely to make acquisitions. The firms' market leverage equals total debt divided by the sum of total debt plus the market value of equity (Harford, Klasa, & Walcott, 2009). Large firms are more likely to make acquisitions than smaller firms (Titman & Wessels, 1988). Bidder size is defined as the natural logarithm of sales (Almazan, De Motta, Titman, & Uysal, 2010). Firms with higher market-to-book (MTB) ratios and stock returns have high likelihoods of acquiring other firms in order to reap the benefits of overvaluation (Uysal, 2011). We use the stock return ratio and the MTB ratio at the fiscal year-end preceding the acquisition announcement date to control for the potential confounding impact of these variables.

To control for industry characteristics, following Schlingemann, Stulz, and Walkling (2002), we employ an industry M&A liquidity index to account for corporate asset liquidity in each industry. We define the industry M&A liquidity index as the total acquisition value, for each year and three-digit SIC code, divided by the total assets of all Compustat firms from the same three-digit SIC code and year. Further, Uysal (2011) finds a negative relationship between industry concentration and the likelihood of acquisitions. Industry concentration is measured by the Herfindahl index, which equals the sum of the squares of sales of a firm divided by the sum of sales of all firms sharing the same three-digit SIC code (Hou & Robinson, 2006). We include industry dummies based on Fama and French's 12-industry classification to account for unobserved time-invariant heterogeneity in industry characteristics. We use year-fixed effects in the regression model to control for unobserved effects over time. We employ random-

effects models to control for unobserved heterogeneity that might be correlated with the likelihood of making an acquisition.<sup>11</sup>

As a further check, we construct another dependent variable that equals the sum of the acquisition value divided by total assets, to address the same relationship as is proposed in H1. Then, we regress this variable on the prior conservatism of the bidder and the same control variables as in model (1) using the random-effects Tobit model, since the dependent variable is censored at zero. We do not use Tobit fixed-effects models because the maximum estimates of these models are biased and Tobit models are non-linear models. Moreover, it is difficult to cluster the standard error in the presence of censored data. Thus, in all the estimations, the robust standard errors are clustered at the firm level, except in the random-effects Tobit models where we use bootstrapping. Bootstrapping is a non-parametric randomisation technique that draws N clusters of observations from firm clusters rather than individual firm-year observations. In our Tobit models we employ 500 bootstrap datasets to estimate standard errors that are robust to heteroscedasticity, correlation, and autocorrelation (Efron, 1979).

Next, for those firms that have engaged in acquisition transactions, we test H2 and H3, which explore the impact of bidder conservatism on target type. We adopt the following probit models to address these issues:

P (Public acquisition=1) =  $\Phi (\beta_0 + \beta_1 \text{ Bidder conservatism }_{i,t-1} + \sum \beta_i \text{ Controls }_{i,t-1})$  (2)

P (DIV2SIC= 1) =  $\Phi \left(\beta_0 + \beta_1 \text{ Bidder conservatism }_{i,t-1} + \sum \beta_i \text{ Controls }_{i,t-1}\right)$  (3)

The dependent variable in model (2) takes the value one if a firm acquires a public target and zero if it acquires a non-public target. In model (3), the dependent variable DIV2SIC is an industrially diversified acquisition dummy that takes the value one if the bidder's two-

<sup>&</sup>lt;sup>11</sup> Due to the small amount of within-firm variation in M&A activity levels, using logit fixed-effects models would exclude more than 50% of our sample observations. In unreported tables, we employ firm-fixed-effects models and standard models and obtain consistent results.

digit SIC code differs from that of the target and zero otherwise (Karampatsas et al., 2014). We further employ a Tobit analysis using the ratio of the acquisition value of each type to the bidders' total assets, since the dependent variable is censored at zero.

We also explore the impact of the conservative ratio on bidders' choice between stock and cash acquisitions. Following Karampatsas et al. (2014), we construct a dependent variable that takes the value one if the deal is financed with more than 50% stock and zero if it is financed with more than 50% cash. We use the following probit model to address H4:

P (Stock dominated=1) =  $\Phi$  ( $\beta_0 + \beta_1$  Bidder conservatism <sub>i,t-1</sub> +  $\sum \beta_i$  Controls <sub>i,t-1</sub>) (4)

We also examine the impact of bidder conservatism on the market reaction to the announcement of acquisition deals. We estimate CARs (cumulative abnormal returns) over a three-day event window (one day before and one day after the announcement date) using a market model. We use two benchmark returns, namely, an equally weighted and a value-weighted index of returns. Then, we use these CARs to run the following fixed-effects ordinary least squares (OLS) regression models to test H5:<sup>12</sup>

CAR <sub>i,t</sub> =  $\beta_0 + \beta_1$  Bidder conservatism <sub>i,t-1</sub> +  $\sum \beta_i$  Controls <sub>i,t-1</sub> +  $\varepsilon_{i,t}$  (5)

Finally, we run all the above models again after replacing the bidder conservatism variable with a highly conservative variable that takes the value one if the conservatism ratio of the bidder is higher than the sample median and zero otherwise. We do so in order to test whether the potential impact of the bidder conservatism ratio highlighted in the previous research questions is identical for highly conservative and less conservative firms.

<sup>&</sup>lt;sup>12</sup> We run fixed-effects OLS to control for endogeneity problems resulting from unobserved heterogeneity of a firm-specific and/or time-invariant nature. In unreported tables, estimating standard OLS and random-effects OLS regressions yields qualitatively similar results.

#### 4. Empirical findings

#### 4.1. Descriptive statistics

Panel A of Table 1 reports descriptive statistics for the dependent variables used in our analysis, as described in the previous section and to be discussed in the following sections. The panel shows that 25.3% (9946=  $0.253\times39260$ ) of US firms, over the sample period of 18 years, make acquisitions, among which 18% (82%) are public (non-public) acquisitions. Around 40% (60%) are industrially diversified (focused) acquisitions. Over the period of the study, on average 24% (76%) of the US firms funded their acquisitions in a stock-dominated (cash-dominated) manner.

Panel B of Table 1 reports the main statistics for the explanatory variables used in our analysis. These descriptive statistics are calculated from a sample of 39,260 firm-year observations. We find that, over the sample period, US non-financial firms' bidder conservatism is on average 1.630, implying the number of negative words relative to total words revealed in the 10-K is 1.63%. The bidder conservatism variable ranges from 0.575 to 2.740, on average, suggesting significant variation in firms' conservative tone over the period of our study. Such variation is further confirmed in Panel C of Table 1, which presents the differences in means for our independent variables, according to whether a firm makes an acquisition. The difference-in-means is tested using the parametric t-statistics. We find that 29,314 firms decide not to make an acquisition, while 9,946 firms make an acquisition during the sample period.

Panel C reports the mean conservative tone for non-acquiring firms as 1.635, while that for acquiring firms is 1.607. The observed difference in conservative tone between nonacquiring and acquiring firms is significant at the 1% level based on a t-test (t= + 4.995), lending preliminary support for our first hypothesis that firms with a more conservative tone would be likely to acquire fewer targets. Panel C reveals that the mean values of market leverage and the Herfindahl index for non-acquiring firms are significantly greater than those for acquiring firms. In contrast, the mean values of other variables, including profitability, bidder size, MTB, stock return, and industry liquidity, for the non-acquiring group are significantly lower than the mean values for the acquiring group.

Panel D reports the differences between highly conservative and less conservative firms in terms of the means and medians of key variables used in our analysis. It reveals that there are significant differences between highly and less conservative firms regarding their profitability, bidder size, stock return, Herfindahl index, and industry liquidity, at the 1% level, based on a t-test. Our results also show that highly conservative firms have significantly lower levels of market leverage (conservative debt levels) and lower levels of MTB than less conservative firms. Our results suggest that highly conservative firms acquire significantly smaller targets (sum of acquisition/TA) than less conservative firms. Accordingly, highly conservative firms might be interested in acquiring these relatively smaller targets so as to manage their risk levels.

#### [Insert Table 1]

#### 4.2. Testing H1: Does bidder conservatism influence the likelihood of making an acquisition?

This section investigates whether bidder conservatism affects a firm's decision to make an acquisition. To address this question, we undertake a probit analysis using a dataset of 39,260 firm-year observations. We also run a Tobit model to test whether bidder conservatism affects the size of an acquisition. Table 2 reports the empirical results for the impact of bidder conservatism on the probability of pursuing an acquisition.

#### [Insert Table 2]

Table 2 reports the coefficient estimates of the probit models in columns 1 and 2 and of the Tobit models in columns 3 and 4. We find that bidder conservatism has a negative association with both the likelihood of engaging in an acquisition and the size of this

acquisition as reported in columns 1 and 3, respectively. These findings support the premise of the behavioural consistency theory, which states that firms conveying a conservative tone in their corporate annual reporting will exhibit the same behaviour in implementing conservative corporate investment policies. Table 2 reports the average marginal effects to assess the economic significance of the bidder conservatism variable. Given that the average marginal effects of bidder conservatism is -0.021 and its standard deviation is 0.442. Thus, a one-standard deviation increase in bidder conservatism variable is related to a -3.67% (= - $0.021\times0.442$ / acquisition dummy sample mean of 0.253) decline in the likelihood of making an acquisition. These findings imply that bidder conservatism seems to have a reasonable economic impact on the likelihood of M&A investments. Our results are also robust to controlling for firm characteristics, industry characteristics, year-fixed effects and unobserved heterogeneity; this provides evidence in support of H1 in that the disclosure tone of 10-K filings could predict firms' subsequent M&A investment decisions.<sup>13</sup>

Further, we examine the impact of bidder conservatism on the probability of making acquisitions after splitting the effect of highly conservative firms from that of less conservative firms. Column 2 of Table 2 shows that highly conservative firms are less likely to make acquisitions. The marginal effects of the variable for highly conservative firms show that it has a non-trivial economic impact. This shows that moving from a less conservative position to a highly conservative one (the change from 0 to 1) reduces the likelihood of an acquisition by 0.013, a decrease of 5% (0.013/ acquisition dummy sample mean of 0.253) over the sample average. Consistent with the lower likelihood of acquisitions, column 4 observes that highly conservative firms invest less in acquisition transactions than do less conservative firms.

<sup>&</sup>lt;sup>13</sup> In unreported analysis, we run standard models and fixed-effects logit models and obtain similar results.

For the other control variables, consistent with Uysal (2011), firms with high leverage ratios are less able to acquire other firms. We find that more profitable firms are more able to make acquisitions. The significant sign of firms' size at the 1% level reveals that bigger firms have a higher probability of acquiring targets than smaller firms. We show that the stock return and the MTB both have significantly positive impacts on the bidder's decision to make an acquisition.

4.3. Testing H2: Does bidder conservatism influence the likelihood of acquiring a public target?

This section examines how a conservative tone affects a firm's decision regarding acquisition type, specifically in terms of public versus non-public acquisitions. Column 1 of Table 3 shows that bidder conservatism has a positive association with the likelihood of undertaking public acquisitions relative to non-public acquisitions. The marginal effects and the standard deviation of the bidder conservatism variable are 0.026 and 0.442, respectively. This indicates that a one-standard deviation increase in bidder conservatism variable is associated with a 6% (0.026× 0.442/ public dummy sample mean of 0.180) larger likelihood of acquiring a public target. These results show that bidder conservatism itself is an economically significant driver of firms' decision over whether to acquire public targets. Further, our findings are in line with the assumption that conservative bidders exhibit greater aversion to ambiguity and uncertainty related to the value of a target. Thus, they prefer public acquisitions in order to avoid the uncertainty and asymmetry of the information associated with non-public acquisitions (Ekkayokkaya et al., 2009). Overall, we accept H2 that there is a positive relationship between a bidder's conservative tone and their likelihood of making a public acquisition.

#### [Insert Table 3]

Next, we test whether the positive relationship between bidder conservatism and the likelihood of acquiring a public target is identical for both highly and less conservative firms. Column 2 of Table 3 reports that highly conservative firms have a higher tendency to acquire public targets than non-public targets. They also invest in public acquisitions more often than less conservative firms, as reported in column 4. Further, the marginal effects confirm the economic impact of the variable for highly conservative firms. Specifically, column 2 reports that moving from a less conservative position to a highly conservative one (the change from 0 to 1) increases the likelihood of acquiring a public target by 0.019.

4.4. Testing H3: Does a bidder's conservatism influence its likelihood of making an industrially diversified acquisition?

This section investigates the relationship between bidder conservatism and firms' acquisition types, in particular, the likelihood of acquiring industrially diverse versus sameindustry targets. Column 1 of Table 4 shows that bidder conservatism reduces the likelihood of making industrially diversified acquisitions. The marginal effects of the bidder conservatism variable indicate that this relation is not only statistically significant, but also economically significant. It suggests that a one-standard deviation increase in bidder conservatism reduces the likelihood of a firm making diversifying acquisitions by 4% (-0.040  $\times$ 0.442/ DIV2SIC sample mean of 0.40). This finding is consistent with the conjuncture that conservative firms are more inclined towards familiar and safe than unfamiliar stimuli. Targets from different industries are viewed as unfamiliar stimuli compared to within-industry acquisitions. Column 3 of Table 4 shows that conservative bidders invest less in industrially diversified acquisitions than in focused acquisitions. This confirms Jost et al.'s (2003) view that conservative bidders prefer focused acquisitions than acquiring targets from industries outside of their own.

#### [Insert Table 4]

Next, we address whether the effect of bidder conservatism on the probability of undertaking industrially diversified acquisitions is systematic for highly versus less conservative firms. Column 2 of Table 4 shows highly conservative firms are less likely to acquire targets from other industries. They also invest less in industrially diversified acquisitions than do less conservative firms, as reported in column 4. The marginal effects of the highly conservative variable also confirm the economic significance of our results.

4.5. Testing H4: Does bidder conservatism influence the M&A payment method?

This section examines the impact of bidder conservatism on the financing method for M&A deals. Table 5 indicates that bidder conservatism affects the payment method of an acquisition. In particular, column 1 finds that bidder conservatism increases the probability of using stock as a medium of payment in acquisition deals. The marginal effects in column 1 suggests that a one-standard deviation increase in bidder conservatism variable is related to a 4% ( $0.023 \times 0.442$  / stock dominated sample mean of 0.237) larger likelihood of making stock acquisitions. Similarly, column 3 shows that bidder conservatism variable increases the percentage of stock in the acquisition transaction. These results are in line with the notion that conservative bidders use stock acquisitions in order to reduce the uncertainty associated with the acquisition process outcomes (Levi & Zhang, 2014). The significance of the results is also robust even when controlling for the bidder's stock return and MTB ratio. This suggests that the impact of bidder conservatism is not derived from either the firms' overpricing effect, or the impact of firms' growth opportunities. However, it confirms H4, in that the conservative tone of a 10-K could predict the method of payment used by a bidder in M&A transactions.

#### [Insert Table 5]

Next, we test whether the effect of bidder conservatism on M&A payment methods is identical for both highly and less conservative firms. Column 2 shows that highly conservative firms prefer stock acquisitions relative to less conservative firms. Column 4 reports that highly

conservative firms use more stock than cash in their acquisition offers. The marginal effects of the highly conservative variable also confirm the economic significance of our results.

For the other control variables, relative size has a significant positive relationship with the likelihood of stock financing at the 1% level. These findings are consistent with Moeller et al.'s (2004) argument that it is difficult to raise sufficient debt for cash offers if the acquisition deal size is larger than the size of the bidder. We also find that bigger firms are more inclined to offer cash for their targets. This relationship is significant at the 1% level. MTB has a positive association with the probability of using stock as a medium of payment for acquisition transactions. Finally, foreign acquisitions are more likely to be financed by cash.

# 4.6. Testing H5: Does bidder conservatism influence abnormal announcement returns around *M&A deals*?

This section investigates whether the conservative tone of prior 10-Ks affects how the market perceives the quality of subsequent acquisition events. Consistent with the market efficiency hypothesis, the market should react accurately and in a timely manner to new information such as acquisition announcements (Fama, 1991). Thus, we use an event study to investigate the immediate impact of acquisition announcements by conservative bidders on shareholders' wealth.

#### [Insert Table 6]

Table 6 presents the impact of bidder conservatism on CARs to the announcements of M&A events. Columns 1 and 3 show that the market reacts negatively when conservative bidder firms announce their M&A deals. In particular, we find that a one-standard deviation increase in bidder conservatism reduces the three-day cumulative abnormal returns around the announcement of an acquisition by approximately 0.0035 (0.008  $\times$  0.442). These results confirm the economic significance of corporate disclosure and its direct impact on how external investors view firms' subsequent investment decisions. Further, columns 2 and 4 show

that highly conservative firms experience more negative stock returns than less conservative firms. Our findings are robust to controlling for firm characteristics, industry characteristics, year-fixed effects, and unobserved heterogeneity, and to using different benchmark returns. This supports H5, that a previous conservative tone could predict the market reaction around subsequent acquisition transactions. It also highlights the mechanism through which bidder sentiment could influence shareholders' wealth.

For the other control variables, Table 6 shows that, consistent with Uysal (2011), highly leveraged firms are likely to be selective and choose the most value-enhancing acquisitions, and thus likely to experience a positive market reaction around M&A events. In line with Moeller et al. (2004) and Uysal (2011), Table 6 finds that the market reacts positively to announcements of acquisitions of targets that are large relative to the acquirer. Further, due to the hubris effect, our results confirm Moller et al.'s (2004) view that the market reacts more negatively to the acquisition announcements of bigger firms than to those of smaller firms. Table 6 documents the market's reaction in favour of M&A deals that are financed totally with cash, and the abnormally poor stock returns experienced by purely stock-financed acquisition deals. These findings are consistent with that of Fuller et al. (2002), in that cash bids usually give a positive signal to the market about the expected future gains from M&A deals. In contrast, the financing of an acquisition entirely with stock might indicate that the bidders are overvalued or the target is a risky investment (Eckbo, 2009).

#### 5. Robustness checks

The foregoing empirical findings section documents that adopting alternative estimation techniques provides robust results. As the previous sections showed, we use different proxies for our dependent variables, i.e., dummy variables, continuous variables as reported in Tables 2 to 5, and different market indices as reported in Table 6. We also estimate our main independent variable using continuous and dummy variables (reported in Tables 2 to

6). Further, we employ different regression models, namely probit and Tobit (see Tables 2 to 5) and OLS (Table 6). We adopt different coefficient estimations: random-effects estimations reported in Tables 2 to 5, fixed-effect estimations reported in Table 6 and standard estimation models, in unreported tables, and obtain similar results. We also run, in unreported tests, fixed-effects estimations for Table 2 and random-effects estimations for Table 6 and again obtain similar findings.

#### 5.1. Multinomial logit for M&A payment method

As a further robustness check, we run a multinomial model in Appendix B to test the effect of bidder conservatism on the payment method used for acquisition transactions. We employ this model since our dependent variable has three different categories, namely pure stock acquisitions, pure cash acquisitions, and mixed payment methods. The variable for pure stock acquisitions takes the value one if the whole transaction is paid for in stock and zero otherwise. Pure cash acquisitions takes the value one if the whole transaction is paid for in cash and zero otherwise. Mixed payment methods takes the value one if the transaction is paid for in both cash and stock and zero otherwise. We choose pure cash acquisitions as our reference group. Column 1 shows that bidder conservatism have a positive association with stock acquisitions, relative to cash acquisitions. Column 2 shows that highly conservative firms are more likely to use their own stocks to finance their M&A deals, relative to less conservative firms. Column 3 finds that bidder conservatism is associated with larger likelihood of using a mix of stock and cash, relative to financing the entire M&A deal with cash only. We do not observe any significant relationship for highly conservative firms in column 4. Overall, the estimates of this multinomial logit model confirm our previous findings that bidder conservatism has a positive association with the likelihood of stock acquisitions.

#### 5.2. Standardised conservatism

Tetlock et al. (2008) use a standardised proxy of the fraction of negative words in each firm-specific news story to account for non-stationary negative tone. This non-stationarity in the negative sentiment might be due to changes in regimes, coverage, or styles of business communication. Following Tetlock et al. (2008), we develop two proxies to control for non-stationarity of conservative tone in a firm's 10-K filings. In particular, we use a standardised conservative words proxy and a standardised bidder conservatism proxy. We estimate the standardised conservative words proxy as the number of negative words in the firm's 10-K minus the mean, divided by the standard deviation, where the mean and standard deviation are calculated at the firm level across the sample period. We similarly define standardised bidder conservatism as the bidder conservatism minus its mean, divided by its standard deviation, again at the firm level. We test the impact of our standardised proxies on the likelihood of making an M&A investment and the consequences of such an investment. **Appendix C** shows, consistent with our previous results, that standardised conservatism has a negative association with the probability of acquiring targets. **Appendix D** supports our previous finding that the market reacts negatively to M&A investments by conservative bidders.

#### 5.3. Bidder conservatism and financial crisis

The collapse of major financial institutions during the financial crisis caused a dramatic fall in global credit markets. This fall led to a severe reduction in the availability of external finance and impeded firms' ability to invest (Duchin, Ozbas, & Sensoy, 2010). Grave, Vardiabasis, and Yavas (2012) find a significant decline in M&A activity during the global financial crisis. Accordingly, it is important to explore the possibility that the negative relationship between bidder conservatism and the likelihood of making an acquisition is driven by the financial crisis shock. We control for the financial crisis effects using three variables, representing the periods prior to, during and post the crisis. The financial crisis variable takes

the value one for the years of the financial crisis, 2008 and 2009, and zero otherwise (Frankel & Saravelos, 2012). The dummy variable prior (post) financial crisis takes the value one for the years before (after) the financial crisis period, and zero otherwise. **Appendix E** shows that the expected relationship between bidder conservatism and the likelihood of making an acquisition still exists and is significantly robust to excluding the effects of the financial crisis and the period subsequent to it.

Further, Samarakoon (2011) reports that, during the financial crisis, the US stock market plummeted by 43%. Thus, it is essential to test whether the negative market reaction around the announcements of M&A deals by conservative bidders is driven by the effect of the financial crisis. **Appendix F** shows that M&A transactions by conservative bidders still experience poor stock returns after we exclude the effects of the financial crisis and the period subsequent to it.

#### 6. Conclusion

This paper provides new insights in the area of information content and the value relevance of one of the major SEC-mandated filings, the 10-K. The paper explores whether the qualitative sections of the 10-K could predict future M&A investment decisions and related movements in the stock market. In particular, it examines the effects of bidders' conservatism or cautionary language in 10-Ks on the likelihood of making an acquisition, the type of the acquisition, the financing choice for the acquisition, and the market reaction around the announcement of the acquisition. According to behavioural consistency theory, we posit that firms that convey a conservative tone in their corporate annual reporting would be likely to exhibit the same behaviour in conducting conservative corporate investment and financing policies. Consistent with this assumption, we find that conservative bidders are conservative in undertaking acquisitions.

Our findings present evidence that the textual sentiment of 10-K reports influences the types of investment made. In particular, we find that conservative bidders prefer public targets, which have less information asymmetry than non-public firms. They also prefer to acquire familiar, that is, within-industry, targets rather than those outside their own industry. According to Levi and Zhang (2014), acquisition deals are risky investments due to the uncertainty associated with their expected outcomes. We document that conservative bidders are more inclined to use their stocks as the medium of financing M&A deals. These results are in line with the notion that conservative bidders use stock acquisition in order to reduce the uncertainty associated with the acquisition process. Finally, we investigate how the market perceives M&A investments made by conservative bidders. We find that these conservative firms experience negative abnormal returns around the announcement of M&A deals. This provides direct evidence of the mechanism through which bidder sentiment could influence shareholders' wealth. Collectively, this paper is one of the first to address the predictive power of mandatory 10-K filings regarding subsequent M&A decisions. Our results have several implications for the area of corporate mandatory disclosure. For instance, investors should carefully analyse corporate textual sentiment in order to effectively grasp corporate attitudes and their potential impact on the quality of future M&A investment decisions.

#### References

Alexandridis, G., Fuller, K. P., Terhaar, L., & Travlos, N. G. (2013). Deal size, acquisition premia and shareholder gains. *Journal of Corporate Finance*, 20, 1-13.

Almazan, A., De Motta, A., Titman, S., & Uysal, V. (2010). Financial structure, acquisition opportunities, and firm locations. *The Journal of Finance*, 65(2), 529-563.

Amihud, Y., DeLong, G. L., & Golubov, A. (2013). Why Bidders Lose. Working paper, New York University.

Ang, J., & Kohers, N. (2001). The take-over market for privately held companies: the US experience. *Cambridge Journal of Economics*, 25(6), 723-748.

Antweiler, W., & Frank, M. Z. (2004). Is all that talk just noise? The information content of internet stock message boards. *The Journal of Finance*, *59*(3), 1259-1294.

Arnold, T., Fishe, R. P., & North, D. (2010). The effects of ambiguous information on initial and subsequent IPO returns. *Financial Management*, *39*(4), 1497-1519.

Baker, M., & Wurgler, J. (2006). Investor sentiment and the cross-section of stock returns. *The Journal of Finance*, *61*(4), 1645-1680.

Balvers, R. J., Gaski, J. F., & McDonald, B. (2015). Financial disclosure and customer satisfaction: Do companies talking the talk actually walk the walk? *Journal of Business Ethics*, *127*(4), 1-17.

Barber, B. M., & Odean, T. (2008). All that glitters: The effect of attention and news on the buying behavior of individual and institutional investors. *Review of Financial Studies*, 21(2), 785-818.

Berger, P. G., & Ofek, E. (1995). Diversification's effect on firm value. *Journal of Financial Economics*, *37*(1), 39-65.

Bharadwaj, A., & Shivdasani, A. (2003). Valuation effects of bank financing in acquisitions. *Journal of Financial Economics*, 67(1), 113-148.

Buehlmaier, M. M., & Whited, T. M. (2014). Looking for Risk in Words: A Narrative Approach to Measuring the Pricing Implications of Financial Constraints. Working paper, University of Rochester.

Capron, L., & Shen, J. C. (2007). Acquisitions of private vs. public firms: Private information, target selection, and acquirer returns. *Strategic Management Journal*, 28(9), 891-911.

Christensen, D. M., Dhaliwal, D. S., Boivie, S., & Graffin, S. D. (2014). Top management conservatism and corporate risk strategies: Evidence from managers' personal political orientation and corporate tax avoidance. *Strategic Management Journal*, available from DOI: 10.1002/smj.2313.

Cronqvist, H., Makhija, A. K., & Yonker, S. E. (2012). Behavioral consistency in corporate finance: CEO personal and corporate leverage. *Journal of Financial Economics*, *103*(1), 20-40.

Custódio, C., & Metzger, D. (2013). How do CEOs matter? The effect of industry expertise on acquisition returns. *Review of Financial Studies*, *26*(6), 2008-2047.

Duchin, R. (2010). Cash holdings and corporate diversification. *The Journal of Finance*, 65(3), 955-992.

Duchin, R., Ozbas, O., & Sensoy, B. A. (2010). Costly external finance, corporate investment, and the subprime mortgage credit crisis. *Journal of Financial Economics*, *97*(3), 418-435.

Eckbo, B. E. (2009). Bidding strategies and takeover premiums: A review. *Journal of Corporate Finance*, 15(1), 149-178.

Efron, B. (1979). Bootstrap methods: another look at the jackknife. *The Annals of Statistics*, 7 (1), 1-26.

Ekkayokkaya, M., Holmes, P., & Paudyal, K. (2009). Limited Information and the Sustainability of Unlisted-Target Acquirers' Returns. *Journal of Business Finance & Accounting*, *36*(9-10), 1201-1227.

Engelberg, J., & Parsons, C. (2011). The causal impact of media in financial markets. *The Journal of Finance*, 66(1), 67-97.

Fama, E. F. (1991). Efficient capital markets: II. The Journal of Finance, 46(5), 1575-1617.

Feldman, R., Govindaraj, S., Livnat, J., & Segal, B. (2010). Management's tone change, post earnings announcement drift and accruals. *Review of Accounting Studies*, *15*(4), 915-953.

Ferris, S. P., Hao, Q., & Liao M. (2013). The Effect of Issuer Conservatism on IPO Pricing and Performance. *Review of Finance*, *17*(3), 993-1027.

Frankel, J., & Saravelos, G. (2012). Can leading indicators assess country vulnerability? Evidence from the 2008–09 global financial crisis. *Journal of International Economics*, 87(2), 216-231.

Fuller, K., Netter, J., & Stegemoller, M. (2002). What do returns to acquiring firms tell us? Evidence from firms that make many acquisitions. *The Journal of Finance*, *57*(4), 1763-1793.

Funder, D. C., & Colvin, C. R. (1991). Explorations in behavioral consistency: properties of persons, situations, and behaviors. *Journal of Personality and Social Psychology*, *60*(5), 773-794.

Garcia, D. (2013). Sentiment during recessions. The Journal of Finance, 68(3), 1267-1300.

Glasgow, M. R., Cartier, A. M., & Wilson, G. D. (1985). Conservatism, sensation-seeking and music preferences. *Personality and Individual Differences*, 6(3), 395-396.

Grave, K., Vardiabasis, D., & Yavas, B. (2012). The global financial crisis and M&A. International Journal of Business and Management, 7(11), 56-66.

Harford, J. (1999). Corporate cash reserves and acquisitions. *The Journal of Finance*, 54(6), 1969-1997.

Harford, J., Klasa, S., & Walcott, N. (2009). Do firms have leverage targets? Evidence from acquisitions. *Journal of Financial Economics*, 93(1), 1-14.

Henry, E. (2008). Are investors influenced by how earnings press releases are written? *Journal of Business Communication*, 45(4), 363-407.

Hou, K., & Robinson, D. T. (2006). Industry concentration and average stock returns. *The Journal of Finance*, *61*(4), 1927-1956.

Huang, A. H., Zang, A. Y., & Zheng, R. (2014). Evidence on the information content of text in analyst reports. *The Accounting Review*, 89(6), 2151-2180.

Hutton, I., Jiang, D., & Kumar, A. (2014). Corporate policies of republican managers. *Journal of Financial and Quantitative Analysis*, 49(5/6), 1279-1310.

Jegadeesh, N., & Wu, D. (2013). Word power: A new approach for content analysis. *Journal of Financial Economics*, *110*(3), 712-729.

Jost, J. T., Glaser, J., Kruglanski, A. W., & Sulloway, F. J. (2003). Political conservatism as motivated social cognition. *Psychological Bulletin*, *129*(3), 339-375.

Kam, C. D., & Simas, E. N. (2010). Risk orientations and policy frames. *The Journal of Politics*, 72(02), 381-396.

Karampatsas, N., Petmezas, D., & Travlos, N. G. (2014). Credit ratings and the choice of payment method in mergers and acquisitions. *Journal of Corporate Finance*, 25, 474-493.

Kearney, C., & Liu, S. (2014). Textual sentiment in finance: A survey of methods and models. *International Review of Financial Analysis*, 33, 171-185.

Lamont, O., & Polk, C. (2001). The diversification discount: Cash flows versus returns, *The Journal of Finance*, *56*(5), 1693-1721.

Lang, L. H., & Stulz, R. M. (1994). Tobin's q, corporate diversification and firm performance. *Journal of Political Economy*, *102*(6), 1248-1280.

Lehavy, R., Li, F., & Merkley, K. (2011). The effect of annual report readability on analyst following and the properties of their earnings forecasts. *The Accounting Review*, 86(3), 1087-1115.

Levi, M., Li, K., & Zhang, F. (2014). Director gender and mergers and acquisitions. *Journal of Corporate Finance*, 28, 185-200.

Li, F. (2010). The information content of forward-looking statements in corporate filings - A naïve Bayesian machine learning approach. *Journal of Accounting Research*, 48(5), 1049-1102.

Lins, K., & Servaes, H. (1999). International evidence on the value of corporate diversification. *The Journal of Finance*, *54*(6), 2215-2239.

Loughran, T., & McDonald, B. (2011). When is a liability not a liability? Textual analysis, dictionaries, and 10-Ks. *The Journal of Finance*, 66(1), 35-65.

Loughran, T., & McDonald, B. (2013). IPO first-day returns, offer price revisions, volatility, and form S-1 language. *Journal of Financial Economics*, *109*(2), 307-326.

Loughran, T., & McDonald, B. (2014). Measuring readability in financial disclosures. *The Journal of Finance*, 69(4), 1643-1671.

Loughran, T., & McDonald, B. (2015a). Textual Analysis in Accounting and Finance: A Survey. Working paper, University of Notre Dame.

Loughran, T., & McDonald, B. (2015b). The use of word lists in textual analysis. *Journal of Behavioral Finance*, *16*(1), 1-11.

Martin, K. J. (1996). The method of payment in corporate acquisitions, investment opportunities, and management ownership. *The Journal of Finance*, *51*(4), 1227-1246.

Matsumoto, D., Pronk, M., & Roelofsen, E. (2011). What makes conference calls useful? The information content of managers' presentations and analysts' discussion sessions. *The Accounting Review*, 86(4), 1383-1414.

Moeller, S. B., Schlingemann, F. P., & Stulz, R. M. (2004). Firm size and the gains from acquisitions. *Journal of Financial Economics*, 73(2), 201-228.

Morck, R., Shleifer, A., & Vishny, R. (1990). Do managerial objectives drive bad acquisitions? *The Journal of Finance*, *45*(1), 31-48.

Officer, M. S. (2007). The price of corporate liquidity: Acquisition discounts for unlisted targets. *Journal of Financial Economics*, 83(3), 571-598.

Officer, M. S., Poulsen, A. B., & Stegemoller, M. (2009). Target-firm information asymmetry and acquirer returns. *Review of Finance*, *13*(3), 467-493.

Reuer, J. J., & Ragozzino, R. (2008). Adverse selection and M&A design: The roles of alliances and IPOs. *Journal of Economic Behavior & Organization*, 66(2), 195-212.

Rogers, J. L., Buskirk, A., & Zechman, S. L. (2011). Disclosure tone and shareholder litigation. *The Accounting Review*, 86(6), 2155-2183.

Samarakoon, L. P. (2011). Stock market interdependence, contagion, and the US financial crisis: The case of emerging and frontier markets. *Journal of International Financial Markets, Institutions & Money*, 21(5), 724-742.

Schlingemann, F. P., Stulz, R. M., & Walkling, R. A. (2002). Divestitures and the liquidity of the market for corporate assets. *Journal of Financial Economics*, *64*(1), 117-144.

Schneider, S. L., & Lopes, L. L. (1986). Reflection in preferences under risk: Who and when may suggest why. *Journal of Experimental Psychology: Human Perception and Performance*, *12*(4), 535-548.

Tetlock, P. C. (2007). Giving content to investor sentiment: The role of media in the stock market. *The Journal of Finance*, *62*(3), 1139-1168.

Tetlock, P. C., Saar-Tsechansky, M., & Macskassy, S. (2008). More than words: Quantifying language to measure firms' fundamentals. *The Journal of Finance*, *63*(3), 1437-1467.

Titman, S., & Wessels, R. (1988). The determinants of capital structure choice. *The Journal of Finance*, 43(1), 1-19.

Uysal, V. B. (2011). Deviation from the target capital structure and acquisition choices. *Journal of Financial Economics*, 102(3), 602-620.

Wilson, G. (2013). *The Psychology of Conservatism* (Routledge Revivals). London: Routledge.

Yan, S. (2015). Managerial attitudes and takeover outcomes: Evidence from corporate filings. *Journal of International Financial Markets, Institutions & Money*, *35*, 30-44.

Yasu, M., & Mukai, A. (2014). Rakuten Drops on \$900 Million Deal to Buy Viber Message App. *Available from BloombergBusiness*.

#### Table 1. Descriptive statistics of the sample

Panel A: Summary statistics of main dependent variables used in subsequent tables

Variables	Ν	Mean	Median	Std. Dev.	Min	Max
Acquisition dummy	39,260	0.253	0.000	0.428	0.000	1.000
Sum of acquisitions/TA	39260	0.087	0.000	0.313	0.000	2.253
Public dummy	9,946	0.180	0.000	0.384	0.000	1.000
Sum of public acquisitions/ TA	9,946	0.108	0.000	0.399	0.000	2.810
DIV2SIC	9,946	0.400	0.000	0.490	0.000	1.000
Diversifying acquisitions/ TA	9,946	0.172	0.000	0.518	0.000	3.676
Stock dominated	9,946	0.237	0.000	0.425	0.000	1.000
% of stock	9,946	0.245	0.000	0.390	0.000	1.000
CAR_equally index	9,701	0.008	0.005	0.048	-0.067	0.094
CAR_value index	9,701	0.008	0.004	0.048	-0.068	0.093
Panel B: Summary statistics o	f full sam	ple				
Variables N	Ме	an	Median S	td. Dev.	Min	Max

variables	N	Mean	Median	Std. Dev.	Min	Max
Bidder conservatism	39,260	1.630	1.632	0.442	0.575	2.740
Market leverage	39,260	0.203	0.122	0.232	0.000	0.934
Profitability	39,260	0.010	0.101	0.355	-2.300	0.395
Bidder size	39,260	5.365	5.433	2.315	0.000	10.693
MTB	39,260	3.048	2.073	5.738	-20.363	37.138
Stock return	39,260	0.204	0.039	0.885	-0.926	5.000
Herfindahl index	39,260	0.140	0.099	0.127	0.031	0.794
Industry M&A liquidity	39,260	0.028	0.014	0.041	0.000	0.214

Panel C: Difference in means between acquiring and non-acquiring firms

	Acquiring firms	Non-acquiring firms	Difference
Variables	Mean	Mean	t-value
Bidder conservatism	1.607	1.635	-4.995***
Market leverage	0.148	0.218	-24.138***
Profitability	0.079	-0.008	19.973***
Bidder size	5.926	5.217	24.841***
MTB	3.733	2.867	12.175***
Stock return	0.370	0.159	19.159***
Herfindahl index	0.131	0.143	-7.737***
Industry M&A liquidity	0.031	0.027	7.734***
Number of observations	9,946	29,314	

	Highly conservative firms	Less conservative firms	<u>Difference</u>
Variables	Mean	Mean	t-value
Market leverage	0.196	0.209	-5.258***
Profitability	-0.022	0.043	-18.268***
Bidder size	5.325	5.404	-3.407***
MTB	2.967	3.128	-2.770***
Sum of acquisitions/ TA	0.082	0.092	-3.212***
Stock return	0.179	0.229	-5.625***
Herfindahl index	0.134	0.147	-10.382***
Industry M&A liquidity	0.027	0.028	1.712***
Number of observations	19,630	19,630	

Panel D: Difference in means between highly and less conservative firms

Panels A, B, C and D of this table give descriptive statistics of all variables as follows: Panel A reports summary statistics of the main dependent variables used in the paper. Panel B reports summary statistics for US public firms with 10-K SEC filings from 1996 to 2013. Panel C reports the t-test on the statistical significance of the difference in means between non-acquiring firms and acquiring firms. Panel D reports t-tests on the statistical significance of the difference in means between highly and less conservative firms. Panels C and D also show the t-values of the differences that are statistically significant at the (\*)10%, (\*\*)5%, and (\*\*\*)1% levels. Variable definitions are given in **Appendix A**. All continuous variables are winsorised at the 1% level.

			(3)	(4)
	(1)	(2)	Sum of	Sum of
	Acquisition	Acquisition	acquisitions/	acquisitions/ TA
	dummy	dummy	TA	
Bidder conservatism	-0.083***		-0.047***	
[Marginal effects]	[-0.021]		[-0.010]	
	(-3.17)		(-2.77)	
Highly conservative				
firms		-0.052**		-0.029**
[Marginal effects]		[-0.013]		[-0.006]
		(-2.40)		(-2.06)
Market leverage	-1.080***	-1.081***	-0.796***	-0.796***
	(-18.60)	(-18.63)	(-19.84)	(-19.86)
Profitability	0.215***	0.222***	0.095***	0.099***
	(4.62)	(4.74)	(3.51)	(3.64)
Bidder size	0.090***	0.089***	0.025***	0.025***
	(12.83)	(12.78)	(5.36)	(5.32)
MTB	0.004**	0.004**	0.004***	0.004***
	(2.44)	(2.45)	(3.90)	(3.91)
Stock return	0.102***	0.102***	0.095***	0.095***
	(9.07)	(9.08)	(13.54)	(13.54)
Herfindahl index	-0.157	-0.164	-0.123	-0.127
	(-1.51)	(-1.57)	(-1.55)	(-1.60)
Industry M&A liquidity	0.064	0.063	-0.056	-0.055
	(0.25)	(0.25)	(-0.33)	(-0.33)
Year FE	Yes	Yes	Yes	Yes
Industry FE	Yes	Yes	Yes	Yes
Ν	39,260	39,260	39,260	39,260

 Table 2. Does bidder conservatism affect the probability of making an acquisition?

This table reports the coefficient estimates of the random-effects probit analysis in columns 1 and 2 and the random-effects Tobit analysis in columns 3 and 4. The dependent variable in the probit models takes the value one if the firm makes an acquisition and zero otherwise. The Tobit analysis estimates the ratio of the sum of the acquisition value to the firm's total assets. Variable definitions are given in **Appendix A**. All continuous variables are winsorised at the 1% level. Marginal effects for bidder conservatism and highly conservative firms are reported in square brackets. *T-statistics* are reported in parenthesis. Standard errors are robust and clustered by firm for the probit models and bootstrapped for the Tobit models. The estimates in the models are statistically significant at the (\*)10%, (\*\*)5%, and (\*\*\*)1% levels.

	(1)	(2)	(3)	(4)
	Public	Public	Sum of public	Sum of public
	dummy	dummy	acquisitions/TA	acquisitions/TA
Bidder Conservatism	0.123***		0.153***	
[Marginal effects]	[0.026]		[0.032]	
	(2.59)		(3.57)	
Highly conservative firms		0.093**		0.097***
[Marginal effects]		[0.019]		[0.020]
		(2.31)		(2.67)
Market leverage	-0.141	-0.142	-0.138	-0.141
	(-1.15)	(-1.16)	(-1.30)	(-1.33)
Profitability	-0.381***	-0.391***	0.074	0.056
	(-2.92)	(-3.00)	(0.66)	(0.51)
Relative size	0.398***	0.397***	0.567***	0.566***
	(16.57)	(16.57)	(32.37)	(32.31)
Bidder size	0.291***	0.292***	0.230***	0.230***
	(22.39)	(22.43)	(18.95)	(18.99)
MTB	0.004	0.004	0.005	0.005
	(1.01)	(1.03)	(1.59)	(1.61)
Stock return	-0.039*	-0.038	-0.017	-0.016
	(-1.66)	(-1.62)	(-0.86)	(-0.81)
Herfindahl index	-0.743***	-0.739***	-0.810***	-0.809***
	(-3.23)	(-3.21)	(-4.22)	(-4.21)
Foreign acquisition	-0.126**	-0.126**	-0.126***	-0.126***
	(-2.57)	(-2.56)	(-3.00)	(-3.00)
Industry M&A liquidity	-1.519***	-1.513***	-1.432***	-1.427***
	(-3.04)	(-3.03)	(-3.17)	(-3.16)
Year FE	Yes	Yes	Yes	Yes
Industry FE	Yes	Yes	Yes	Yes
Ν	9,946	9,946	9,946	9,946

Table 3. Does bidder conservatism affect the probability of making public versus nonpublic acquisitions?

This table reports the coefficient estimates of the random-effects probit analysis in columns 1 and 2 and the random-effects Tobit analysis in columns 3 and 4. The dependent variable in the probit models takes the value one if the firm acquires a public target and zero if it acquires a non-public target. Tobit analysis estimates the ratio of the sum of public acquisition values to the firm's total assets. Variable definitions are given in **Appendix A**. All continuous variables are winsorised at the 1% level. Marginal effects for bidder conservatism and highly conservative firms are reported in square brackets. *T*-statistics are reported in parenthesis. Standard errors are robust and clustered by firm for the probit models and bootstrapped for the Tobit models. The estimates in the models are statistically significant at the (\*)10%, (\*\*)5%, and (\*\*\*)1% levels.

<b>. 1</b>			(3)	(4)
	(1)	(2)	Diversifying	Diversifying
	DIV2SIC	DIV2SIC	acquisitions/ TA	acquisitions/ TA
Bidder conservatism	-0.105**		-0.043*	
[Marginal effects]	[-0.040]		[-0.013]	
	(-2.55)		(-1.79)	
Highly conservative firms		-0.120***		-0.062***
[Marginal effects]		[-0.045]		[-0.018]
		(-3.45)		(-2.98)
Market leverage	-0.189*	-0.195*	-0.166***	-0.172***
	(-1.72)	(-1.77)	(-2.91)	(-3.02)
Profitability	-0.431***	-0.433***	-0.345***	-0.350***
	(-4.05)	(-4.06)	(-5.48)	(-5.59)
Relative size	-0.009	-0.009	0.317***	0.317***
	(-0.52)	(-0.52)	(27.25)	(27.25)
Bidder size	0.059***	0.058***	0.017***	0.017***
	(3.92)	(3.90)	(3.13)	(3.09)
MTB	-0.006	-0.006	-0.003*	-0.004*
	(-1.52)	(-1.56)	(-1.75)	(-1.78)
Stock return	-0.024	-0.024	0.017	0.017
	(-1.26)	(-1.31)	(1.50)	(1.47)
Herfindahl index	1.457***	1.449***	0.794***	0.786***
	(8.73)	(8.68)	(9.03)	(8.95)
Foreign acquisition	0.003	0.003	-0.008	-0.008
	(0.07)	(0.07)	(-0.29)	(-0.29)
Industry M&A liquidity	-0.203	-0.192	-0.424	-0.412
	(-0.50)	(-0.47)	(-1.57)	(-1.53)
Year FE	Yes	Yes	Yes	Yes
Ν	9,946	9,946	9,946	9,946

Table 4. Does bidder conservatism affect the probability of the firm making an industrially diversified acquisition versus a same-industry acquisition?

This table reports the coefficient estimates of a random-effects probit analysis in columns 1 and 2 and a random-effects Tobit analysis in columns 3 and 4. The dependent variable in the probit models takes the value one if the bidder's two-digit SIC code differs from the target's two-digit SIC code and zero otherwise. The Tobit analysis estimates the ratio of the sum of the values of diversifying acquisitions to the firm's total assets. Variable definitions are given in **Appendix A**. All continuous variables are winsorised at the 1% level. Marginal effects for bidder conservatism and highly conservative firms are reported in square brackets. *T*-statistics are reported in parenthesis. Standard errors are robust and clustered by firm for the probit models and bootstrapped for the Tobit models. The estimates in the models are statistically significant at the (\*)10%, (\*\*)5%, and \*\*\*1% levels. We do not include industry-fixed effects in these models because we wish to observe the impact of bidder conservatism on our main variable of interest, the dependent variable, which is industrially diversified acquisitions.

		1 5		
	(1)	(2)	(3)	(4)
	Stock dominated	Stock dominated	% of stock	% of stock
Bidder conservatism	0.116**		0.064***	
[Marginal effects]	[0.023]		[0.020]	
	(2.20)		(2.68)	
Highly conservative firms		0.076*		0.041**
[Marginal effects]		[0.015]		[0.013]
		(1.69)		(1.97)
Market leverage	-0.050	-0.065	-0.008	-0.010
	(-0.37)	(-0.49)	(-0.14)	(-0.16)
Profitability	-1.254***	-1.250***	-0.439***	-0.446***
	(-7.99)	(-8.01)	(-7.66)	(-7.80)
Relative size	0.323***	0.322***	0.138***	0.138***
	(9.70)	(9.62)	(14.11)	(14.08)
Bidder size	-0.059***	-0.065***	-0.067***	-0.067***
	(-4.23)	(-4.69)	(-10.48)	(-10.42)
MTB	0.032***	0.032***	0.015***	0.015***
	(6.70)	(6.70)	(8.54)	(8.55)
Stock return	0.016	0.017	0.012	0.012
	(0.73)	(0.77)	(1.25)	(1.27)
Herfindahl index	-0.758***	-0.769***	-0.303***	-0.300***
	(-3.06)	(-3.10)	(-2.73)	(-2.71)
Foreign acquisition	-0.360***	-0.358***	-0.208***	-0.208***
	(-6.67)	(-6.62)	(-8.18)	(-8.17)
Industry M&A liquidity	0.863	0.871	0.295	0.301
	(1.62)	(1.64)	(1.19)	(1.22)
Year FE	Yes	Yes	Yes	Yes
Industry FE	Yes	Yes	Yes	Yes
Ν	9,946	9,946	9,946	9,946

Table 5. Does bidder conservatism affect the payment method used for M&As?

This table reports the coefficient estimates of random-effects probit analysis in columns 1 and 2 and random-effects Tobit analysis in columns 3 and 4. The dependent variable in the probit models takes the value one if the deal is financed with more than 50% stock and zero if it is financed with more than 50% cash. The Tobit analysis estimates the percentage of stock financing in the M&A transactions. Variable definitions are given in **Appendix A**. All continuous variables are winsorised at the 1% level. Marginal effects for bidder conservatism and highly conservative firms are reported in square brackets. *T*-statistics are reported in parenthesis. Standard errors are robust and clustered by firm for the probit models and bootstrapped for the Tobit models. The estimates in the models are statistically significant at the (\*)10%, (\*\*)5%, and (\*\*\*)1% levels.

	(1)	(2)	(3)	(4)
	CAR_equally	CAR_equally	CAR_value	CAR_valu
	index	index	index	e index
Bidder conservatism	-0.008*		-0.007*	
	(-1.85)		(-1.71)	
Highly conservative firms		-0.005*		-0.005*
		(-1.70)		(-1.68)
Market leverage	0.027*	0.028*	0.025*	0.025*
-	(1.93)	(1.95)	(1.78)	(1.80)
Profitability	-0.005	-0.005	-0.006	-0.006
	(-0.52)	(-0.48)	(-0.63)	(-0.60)
Relative size	0.009**	0.009*	0.011**	0.011**
	(1.98)	(1.95)	(2.33)	(2.30)
Bidder size	-0.008***	-0.008***	-0.008***	-0.008***
	(-3.46)	(-3.41)	(-3.37)	(-3.32)
MTB	-0.002***	-0.002***	-0.002***	-0.002***
	(-2.64)	(-2.64)	(-2.65)	(-2.66)
Stock return	0.001	0.001	0.001	0.001
	(0.69)	(0.69)	(0.83)	(0.83)
Herfindahl index	0.017	0.018	0.019	0.020
	(0.60)	(0.61)	(0.67)	(0.68)
Foreign acquisition	-0.001	-0.001	-0.001	-0.001
	(-0.45)	(-0.45)	(-0.30)	(-0.30)
Industry M&A liquidity	-0.071	-0.071	-0.063	-0.062
	(-1.34)	(-1.34)	(-1.19)	(-1.19)
Competed dummy	0.001	0.001	-0.000	0.000
	(0.09)	(0.10)	(-0.01)	(0.00)
Hostile dummy	-0.021	-0.021	-0.019	-0.019
	(-1.06)	(-1.05)	(-0.96)	(-0.95)
Pure cash acquisition	0.005**	0.005**	0.004	0.004
	(2.05)	(2.07)	(1.52)	(1.54)
Pure stock acquisition	-0.009**	-0.009**	-0.009***	-0.009***
	(-2.44)	(-2.44)	(-2.68)	(-2.68)
Year FE	Yes	Yes	Yes	Yes
Firm FE	Yes	Yes	Yes	Yes
Ν	9,701	9,701	9,701	9,701

 Table 6. Does bidder conservatism affect the market reaction to M&A deals?

This table reports the coefficient estimates of fixed-effects OLS analyses. The dependent variable is the CAR (cumulative abnormal return) estimated over a three-day event window (from one day before to one day after the announcement date). For columns 1 and 2, the benchmark return is the equally weighted index of returns, and for columns 3 and 4 it is the value-weighted index of returns including dividends for the New York Stock Exchange, American Stock Exchange, and NASDAQ combined. Variable definitions are given in **Appendix A**. All continuous variables are winsorised at the 1% level. The standardised betas for bidder conservatism and highly conservative firms are reported in square brackets. *T*-statistics are reported in parenthesis. Standard errors are robust and clustered by acquiring

firms. The estimates in the models are statistically significant at the (\*)10%, (\*\*)5%, and (\*\*\*)1% levels.

Appendix A. variable definitions	Appendix A	. Variable	definitions
----------------------------------	------------	------------	-------------

Variable	Definition	Unit
Acquisition dummy	Takes the value one if the firm makes an acquisition and zero otherwise.	0, 1
Bidder conservatism	Frequency of negative words divided by total number of words in the 10-K, multiplied by 100, using Loughran and McDonald's (2011) word list.	%
Bidder size	Natural logarithm of sales (\$million).	LN
CAR_equally index	Cumulative abnormal returns estimated over a three-day event window (from one day before to one day after the announcement date). The benchmark return is the equally weighted index of returns.	%
CAR_value index	Cumulative abnormal returns estimated over a three-day event window (from one day before to one day after the announcement date). The benchmark return is the value- weighted index of returns, including dividends, for the New York Stock Exchange, American Stock Exchange, and NASDAQ combined.	%
Competed dummy	Takes the value one if there is more than one bidder and zero otherwise.	0, 1
DIV2SIC	Dummy variable that takes the value one if the bidder's two-digit SIC code differs from that of the target, and zero otherwise.	0, 1
Diversifying acquisitions/ TA	The ratio of the sum of diversifying acquisition values (\$million) to the firm's total assets (\$million).	0, 1
During financial crisis	Dummy variable that takes the value one for the years 2008 and 2009 and zero otherwise.	0, 1
During crisis* bidder conservatism	An interaction term between the during financial crisis and bidder conservatism variables.	%
Foreign acquisition	Dummy variable that takes the value one if a firm acquires a foreign target and zero otherwise.	0, 1
Herfindahl index	The sum of the squares of sales (\$million) of a firm divided by the sum of sales (\$million) of all firms sharing the same three-digit SIC code.	%
Highly conservative firms	Dummy variable that takes the value one if the bidder conservatism ratio for the firm is higher than the sample median and zero otherwise.	0, 1
Hostile dummy	Takes the value one if the deal is a hostile acquisition and zero otherwise.	0, 1
Industry M&A liquidity	Total acquisition value (\$million), for each year and three-digit SIC code, scaled by the total assets (\$million) of all US firms that share the same three-digit SIC code and year.	%
Market leverage	Total debt (\$million) divided by the sum of total debt plus the market value of equity (\$million).	%

Mixed payments method	Dummy variable that takes the value one if the acquisition transaction is paid in both cash and stock, and zero otherwise.	0, 1
MTB	Market value (\$million) over the book value of total assets (\$million).	%
% of stock	Percentage of stock financing in the M&A transactions.	%
Post financial crisis	Dummy variable that takes the value one for years after 2009 and zero otherwise.	0, 1
Post crisis* bidder conservatism	An interaction term between the post financial crisis and bidder conservatism variables.	%
Profitability	Earnings before interest, taxes and depreciations (\$million) divided by total assets (\$million).	%
Public dummy	Takes the value one if the firm acquires a public target and zero if it acquires a non-public target.	0, 1
Pure cash acquisition	Dummy variable that takes the value one if the whole transaction is paid for in cash and zero otherwise.	0, 1
Pure stock acquisition	Dummy variable that takes the value one if the whole transaction is paid for in stock and zero otherwise.	0, 1
Relative size	Natural logarithm of the ratio of the deal value (\$million) to the acquirers' total assets (\$million) prior to the announcement date.	LN
Stand. bidder conservatism	Bidder conservatism ratio minus its mean divided by its standard deviation, at the firm level.	%
Stand. conservative words	The number of negative words minus the mean, divided by the standard deviation, at the firm level.	%
Stock dominated	Dummy variable that takes the value one if the deal is financed with more than 50% stock and zero if financed with more than 50% cash.	0, 1
Sum of acquisitions/ TA	The ratio of the sum of the acquisition values (\$million) to the firm's total assets (\$million).	%
Sum of public acquisitions/ TA	The ratio of the sum of public acquisition values (\$million) to the firm's total assets (\$million).	%
Stock return	The compounded total stock returns one year prior to a firm's fiscal year-end.	%

	(1)	(2)	(3)	(4)
	Pure stock vs.	Pure stock vs.	Mixed payment	Mixed payment
	pure cash	pure cash	vs. pure cash	vs. pure cash
Bidder conservatism	0.215**		0.151**	
	(2.28)		(2.14)	
Highly conservative firms		0.138*		0.068
		(1.66)		(1.17)
Market leverage	-0.767***	-0.788***	0.189	0.167
	(-2.62)	(-2.65)	(1.21)	(1.07)
Profitability	-2.906***	-2.961***	-1.351***	-1.407***
	(-8.68)	(-8.88)	(-4.72)	(-4.92)
Relative size	1.137***	1.137***	0.979***	0.980***
	(8.27)	(8.27)	(7.37)	(7.37)
Bidder size	-0.090***	-0.088***	-0.173***	-0.172***
	(-2.66)	(-2.60)	(-9.78)	(-9.70)
MTB	0.057***	0.058***	0.009	0.009
	(5.20)	(5.20)	(1.25)	(1.28)
Stock return	0.057	0.058	0.070**	0.071**
	(1.36)	(1.38)	(2.13)	(2.16)
Herfindahl index	-1.886***	-1.893***	-0.115	-0.128
	(-4.38)	(-4.39)	(-0.52)	(-0.58)
Foreign acquisition	-0.515***	-0.514***	-0.206***	-0.206***
	(-4.96)	(-4.95)	(-3.14)	(-3.14)
Industry M&A liquidity	0.458	0.479	-0.265	-0.244
	(0.75)	(0.78)	(-0.46)	(-0.42)
Year FE	Yes	Yes	Yes	Yes
Industry FE	Yes	Yes	Yes	Yes
Ν	9,946	9,946	9,946	9,946

Appendix B. Robustness check of the impact of bidder conservatism on the payment method for M&As

This table reports the coefficient estimates of the multinomial-logit analysis. The dependent variable has three different categories, namely pure stock acquisitions, pure cash acquisitions and mixed payment methods. Pure stock acquisitions takes the value one if the whole transaction is paid for in stock and zero otherwise. Pure cash acquisitions takes the value one if the whole transaction is paid for in cash and zero otherwise. Mixed payments method takes the value one if the transaction is paid for in both cash and stock and zero otherwise. The estimated coefficients in this table report differences relative to the baseline group of pure cash acquisitions relative to the baseline group. Columns 3 and 4 report the coefficient estimates for mixed payment methods relative to the baseline group. Variable definitions are given in Appendix A. All continuous variables are winsorised at the 1% level. *T*-statistics are reported in parenthesis. Standard errors are robust and clustered by acquiring firms. The estimates in the models are statistically significant at the (\*)10%, (\*\*)5%, and (\*\*\*)1% levels.

ε τη				
	(1)	(2)	(3)	(4)
	Acquisition	Acquisition	Sum of	Sum of
	dummy	dummy	acquisitions/ TA	acquisitions/ TA
Stand. conservative words	-0.053***		-0.035***	
	(-4.78)		(-5.10)	
Stand. bidder conservatism		-0.054***		-0.044***
		(-4.91)		(-6.35)
Market leverage	-1.064***	-1.066***	-0.783***	-0.781***
	(-18.28)	(-18.40)	(-19.52)	(-19.52)
Profitability	0.212***	0.217***	0.091***	0.093***
	(4.56)	(4.65)	(3.39)	(3.46)
Bidder size	0.090***	0.090***	0.025***	0.025***
	(12.86)	(12.88)	(5.34)	(5.36)
MTB	0.004**	0.004**	0.004***	0.004***
	(2.46)	(2.42)	(3.92)	(3.86)
Stock return	0.102***	0.102***	0.094***	0.094***
	(8.99)	(9.03)	(13.44)	(13.47)
Herfindahl index	-0.169	-0.170	-0.130*	-0.129
	(-1.61)	(-1.63)	(-1.65)	(-1.63)
Industry M&A liquidity	0.090	0.099	-0.037	-0.026
	(0.36)	(0.39)	(-0.22)	(-0.15)
Year FE	Yes	Yes	Yes	Yes
Industry FE	Yes	Yes	Yes	Yes
Ν	39,209	39,209	39,209	39,209

Appendix C. Robustness check of the	effect of	f the standardise	d conservatism	on	the
likelihood of undertaking acquisitions					

This table reports the coefficient estimates of random-effects probit analysis in columns 1 and 2 and random-effects Tobit analysis in columns 3 and 4. The dependent variable in the probit models takes the value one if the firm makes an acquisition. The Tobit analysis estimates the ratio of the sum of all the acquisition values to the firm's total assets. Variable definitions are given in Appendix A. All continuous variables are winsorised at the 1% level. *T-statistics* are reported in parenthesis. Standard errors are robust and clustered by firm for the probit models and bootstrapped for the Tobit models. The estimates in the models are statistically significant at the (\*)10%, (\*\*)5%, and (\*\*\*)1% levels.

1				
	(1)	(2)	(3)	(4)
	CAR_equally	CAR_equally	CAR_value	CAR_value
	index	index	index	index
Stand. conservative words	-0.003*		-0.003*	
	(-1.91)		(-1.81)	
Stand. bidder conservatism		-0.002*		-0.002**
		(-1.82)		(-2.14)
Market leverage	0.014	0.018**	0.010	0.017**
	(0.85)	(2.17)	(0.65)	(2.02)
Profitability	-0.006	-0.008	-0.008	-0.007
	(-0.56)	(-1.30)	(-0.70)	(-1.13)
Relative size	0.005	0.006**	0.006	0.008***
	(0.98)	(2.31)	(1.14)	(2.82)
Bidder size	-0.010***	-0.005***	-0.010***	-0.005***
	(-3.63)	(-3.50)	(-3.56)	(-3.38)
MTB	-0.002**	0.000	-0.002**	0.000
	(-2.22)	(0.52)	(-2.29)	(0.49)
Stock return	0.001	-0.001	0.001	-0.001
	(0.78)	(-0.96)	(0.87)	(-1.07)
Herfindahl index	0.043	-0.004	0.047	0.000
	(1.31)	(-0.26)	(1.47)	(0.01)
Foreign acquisition	0.000	-0.000	0.001	-0.001
	(0.07)	(-0.30)	(0.27)	(-0.42)
Industry M&A liquidity	-0.070	-0.033	-0.063	-0.030
	(-1.17)	(-1.06)	(-1.06)	(-0.97)
Competed dummy	0.006	0.005	0.005	0.004
	(0.52)	(0.96)	(0.46)	(0.70)
Hostile dummy	-0.013	-0.016	-0.011	-0.014
	(-0.58)	(-1.36)	(-0.46)	(-1.21)
Pure cash acquisition	0.006**	0.004**	0.004	0.003**
	(2.05)	(2.53)	(1.47)	(1.96)
Pure stock acquisition	-0.008**	-0.008***	-0.009**	-0.008***
	(-2.02)	(-3.85)	(-2.27)	(-3.99)
Year FE	Yes	Yes	Yes	Yes
Firm FE	Yes	Yes	Yes	Yes
Ν	9,691	9,691	9,691	9,691

Appendix D. Robustness check of the effect of the standardised conservatism on acquirers' cumulative abnormal returns

This table reports the coefficient estimates of fixed-effects OLS analyses. The dependent variable is CARs (cumulative abnormal returns) estimated over a three-day event window (from one day before to one day after the announcement date). The benchmark return in columns 1 and 2 is the equally weighted index of returns and in columns 3 and 4 is the value-weighted index of returns, including dividends, for the New York Stock Exchange, American Stock Exchange, and NASDAQ combined. Variable definitions are given in Appendix A. All continuous variables are winsorised at the 1% level. *T*-statistics are reported in parenthesis. Standard errors are robust and clustered by acquiring firms. The estimates in the models are statistically significant at the (\*)10%, (\*\*)5%, and (\*\*\*)1% levels.

			(3)	(4)
	(1)	(2)	Sum of	Sum of
	Acquisition	Acquisition	acquisitions/	acquisitions/ TA
	dummy	dummy	TA	
Bidder conservatism	-0.163***	-0.166***	-0.130***	-0.133***
	(-6.47)	(-6.27)	(-7.95)	(-7.74)
Market leverage	-1.103***	-1.103***	-0.821***	-0.821***
	(-19.20)	(-19.20)	(-20.38)	(-20.38)
Profitability	0.272***	0.271***	0.153***	0.153***
	(5.70)	(5.68)	(5.56)	(5.53)
Bidder size	0.074***	0.075***	0.010**	0.010**
	(10.41)	(10.42)	(2.08)	(2.09)
MTB	0.005***	0.005***	0.005***	0.005***
	(2.68)	(2.66)	(4.53)	(4.52)
Stock return	0.100***	0.100***	0.093***	0.093***
	(9.36)	(9.36)	(13.69)	(13.69)
Herfindahl index	-0.234**	-0.234**	-0.203**	-0.203**
	(-2.23)	(-2.23)	(-2.50)	(-2.51)
Industry M&A liquidity	0.105	0.107	0.099	0.101
	(0.45)	(0.46)	(0.63)	(0.64)
During financial crisis	-0.170***	-0.120	-0.144***	-0.102
	(-5.23)	(-0.77)	(-6.46)	(-0.94)
Post financial crisis	-0.078***	-0.183	-0.045**	-0.140
	(-2.58)	(-1.25)	(-2.33)	(-1.48)
During crisis*bidder conservatism		-0.027		-0.023
		(-0.32)		(-0.39)
Post crisis* bidder conservatism		0.058		0.053
		(0.73)		(1.02)
Industry FE	Yes	Yes	Yes	Yes
Ν	39,260	39,260	39,260	39,260

Appendix E. Robustness check of the effect of bidder conservatism on the likelihood of undertaking acquisitions during the financial crisis

This table reports the coefficient estimates of random-effects probit analysis in columns 1 and 2 and random-effects Tobit analysis in columns 3 and 4. The dependent variable in the probit models takes the value one if the firm makes an acquisition. Tobit analysis estimates the ratio of the sum of the acquisition values to the firm's total assets. Variable definitions are given in Appendix A. All continuous variables are winsorised at the 1% level. *T-statistics* are reported in parenthesis. Standard errors are robust and clustered by firm for the probit models and bootstrapped for the Tobit models. The estimates in the models are statistically significant at the (\*)10%, (\*\*)5%, and (\*\*\*)1% levels.

	8			
	(1)	(2)	(3)	(4)
	CAR_equally	CAR_equally	CAR_value	CAR_value
	index	index	index	index
Bidder conservatism	-0.008**	-0.008**	-0.008**	-0.008*
	(-2.09)	(-2.03)	(-1.98)	(-1.95)
Market leverage	0.027**	0.027**	0.025*	0.025*
	(1.97)	(1.96)	(1.79)	(1.78)
Profitability	-0.004	-0.004	-0.006	-0.006
	(-0.41)	(-0.41)	(-0.63)	(-0.63)
Relative size	0.009**	0.009**	0.011**	0.011**
	(2.01)	(2.00)	(2.36)	(2.35)
Bidder size	-0.009***	-0.009***	-0.009***	-0.009***
	(-4.59)	(-4.57)	(-4.46)	(-4.44)
MTB	-0.002***	-0.002***	-0.002**	-0.002**
	(-2.60)	(-2.61)	(-2.57)	(-2.57)
Stock return	0.000	0.000	0.001	0.001
	(0.49)	(0.49)	(0.94)	(0.93)
Herfindahl index	0.016	0.016	0.017	0.017
	(0.56)	(0.55)	(0.61)	(0.60)
Foreign acquisition	-0.001	-0.001	-0.001	-0.001
	(-0.45)	(-0.45)	(-0.30)	(-0.30)
Industry M&A liquidity	-0.082*	-0.082*	-0.063	-0.062
	(-1.68)	(-1.66)	(-1.30)	(-1.29)
Competed dummy	0.000	0.000	-0.000	-0.000
	(0.05)	(0.05)	(-0.05)	(-0.05)
Hostile dummy	-0.020	-0.020	-0.019	-0.018
	(-0.99)	(-0.99)	(-0.94)	(-0.93)
Pure cash acquisition	0.005**	0.005**	0.004	0.004
	(2.05)	(2.05)	(1.53)	(1.52)
Pure stock acquisition	-0.008**	-0.008**	-0.009***	-0.009***
	(-2.34)	(-2.35)	(-2.61)	(-2.62)
During financial crisis	0.004	0.009	0.004	0.008
	(0.96)	(0.44)	(0.95)	(0.36)
Post financial crisis	0.012***	0.008	0.011***	0.005
	(3.21)	(0.41)	(3.12)	(0.28)
During crisis*bidder conservatism		-0.003		-0.002
-		(-0.27)		(-0.19)
Post crisis* bidder conservatism		0.002		0.003
		(0.20)		(0.31)
Firm FE	Yes	Yes	Yes	Yes
Ν	9,691	9,691	9,691	9,691

Appendix F. Robustness check of the effect of bidder conservatism on the acquirers' cumulative abnormal returns during the financial crisis

This table reports the coefficient estimates of fixed-effects OLS analyses. The dependent variable is the CAR estimated over a three-day event window. The benchmark return in columns 1 and 2 is the equally weighted index of returns and in columns 3 and 4 is the value-weighted index of returns. Variable definitions are given in **Appendix A**. All continuous

variables are winsorised at the 1% level. *T*-statistics are reported in parenthesis. Standard errors are robust and clustered by acquiring firms. The estimates in the models are statistically significant at the (\*)10%, (\*\*)5%, and (\*\*\*)1% levels.